Author’s response to reviews

Title: The Efficacy of Continuous versus Single-injection Femoral Nerve Block in Total Knee Arthroplasty: A Systematic Review and Meta-analysis

Authors:

Hsuan-Hsiao Ma (tyc20192@gmail.com)
Te-Feng Arthur Chou (arthrouchou2@gmail.com)
Shang-Wen Tsai (ray366kimo@gmail.com)
Cheng-Fong Chen (swtsai.vghtpe@gmail.com)
Po-Kuei Wu (drwuvgh@gmail.com)
Wei-Ming Chen (wmchen2499@gmail.com)

Version: 4 Date: 16 Feb 2020

Author’s response to reviews:

Thank you for all the efforts to review our manuscript. As your concerns of overlap with the previous published work, we have revised our manuscripts especially in methods and results. We have cited the Ma, H-H. et al. (2019) "The efficacy of intra-operative periarticular injection in Total hip arthroplasty: a systematic review and meta-analysis" as one of the reference in the method section. Although some of the idea, research strategy and the presenting results parameter were similar to the aforementioned article, we ensured that the statement of the text was revised and reformulated.